

John Kolb John Yang Randy H. Katz David E. Culler

Electrical Engineering and Computer Sciences University of California at Berkeley

Technical Report No. UCB/EECS-2020-178 http://www2.eecs.berkeley.edu/Pubs/TechRpts/2020/EECS-2020-178.html

August 31, 2020

Copyright © 2020, by the author(s). All rights reserved.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. To copy otherwise, to republish, to post on servers or to redistribute to lists, requires prior specific permission.

John Kolb Computer Science Division University of California, Berkeley jkolb@cs.berkeley.edu

Randy H. Katz Computer Science Division University of California, Berkeley randy@cs.berkeley.edu

ABSTRACT

We present *Quartz*, a language and framework for developing and validating smart contracts. A user writes a *Quartz* contract as an extended finite-state machine in a domain-specific language. *Quartz* translates this description into a formal specification in TLA⁺. The specification captures both the contract's logic and the execution semantics of the blockchain. *Quartz* uses bounded model checking to automatically determine if a contract adheres to user-defined properties. Once validated, the contract's author uses *Quartz* to generate a Solidity implementation for deployment on a blockchain.

We used a survey of 16 contract case studies, drawn from a variety of domains, to motivate the design of *Quartz*'s DSL and to evaluate its effectiveness. We show that *Quartz* contracts are consistently more concise than handwritten Solidity equivalents. We present two in-depth case studies where *Quartz* identifies multiple significant contract vulnerabilities and provides useful execution traces to assist the developer in addressing them. Finally, we demonstrate that *Quartz* imposes only modest overhead in terms of generated code size and execution efficiency over handwritten Solidity. *Quartz* is able to automatically verify functional, contract-specific properties. It is the first tool of this type that detects vulnerabilities arising from interactions with external code, such as reentrancy.

1 INTRODUCTION

Recent developments in blockchains have generated renewed interest in distributed ledger systems. Originally conceived as an immutable and tamper-proof record of financial transactions, blockchains have expanded into the role of secure and robust data repositories for general-purpose applications with the advent of *smart contracts*. A smart contract defines a body of data and the logic for a set of transformations that may be applied to this data. A contract's initial state and all subsequent transformations are recorded on a distributed ledger, allowing any consumer of the ledger to inspect the current state of the contract and to validate the history of operations that produced this state. Proper operation of the ledger is ensured by a protocol leveraging cryptographic primitives in which each participant acts out of self interest.

A smart contract operates under the same assumptions — it defines a protocol for multiple distrusting, self-interested parties to execute transactions against a shared body of state. Users rely on the underlying ledger to enact this protocol by faithfully executing

John B. Yang Computer Science Division University of California, Berkeley john.yang20@berkeley.edu

David E. Culler Computer Science Division University of California, Berkeley culler@cs.berkeley.edu



Figure 1: Quartz Architecture

the contract's logic precisely as written. It is therefore up to the contract's code to ensure the integrity of its data and to enforce the application-level guarantees its users expect. The correctness and security of the contract's code is therefore a central concern for its developers and users.

Smart contracts offer powerful capabilities but have proven difficult to engineer for security and correctness. Serious vulnerabilities have been discovered in smart contracts deployed in production systems, even those written by domain experts [44]. The execution semantics for contract code have significant differences from those of more traditional software, making it difficult to identity potential vulnerabilities. A contract's code is immutable once deployed to a ledger, meaning issues found only after release cannot be patched. Moreover, the internals of any deployed smart contract are open to all of a ledger's users, who are also free to invoke arbitrary transactions against the contract at any time. Anyone may therefore inspect a contract for vulnerabilities and then attempt to exploit them. There is significant incentive to do so because contracts are often used to manage sensitive data and track valuable assets.

Given these concerns, there is significant value in representing a contract clearly and concisely. Second, a contract's authors would like to understand and verify as much as possible about its potential behaviors and vulnerabilities before it is deployed to the open environment of a distributed ledger. Third, the authors need the ability to generate a working implementation ready for use from their concise contract description.

This paper presents *Quartz*, a framework for engineering secure smart contracts that addresses these three goals. The design of this framework is shown in Figure 1. *Quartz* offers a domain-specific

language in which a developer describes her contract as an extended finite-state machine. The transitions of this state machine become the transactions against the contract recorded on the underlying ledger. The contract author augments her description with a set of invariants regarding the contract's behavior. *Quartz* translates a state machine and invariants into a formal specification suitable for model checking, expressed using TLA⁺ [34]. A model checker then searches possible contract execution paths for violations of the stated invariants. If a violation is found, the model checker presents

the contract author with an execution trace that induces it.

Once the contract obeys the checked invariants, possibly after an iterative process of refining the contract and rerunning the model checker, the author may use *Quartz* to generate a Solidity [18] implementation from the original state machine description. This allows her to seamlessly deploy the contract to any Ethereumbacked blockchain, with greater assurance that what is deployed in production will behave as expected.

Quartz leverages its smart contract description language to make smart contracts both easier to express, facilitating contract development, and easier to analyze, facilitating systematic and robust contract testing. The design of this DSL is informed by a survey of 16 case studies based on real contracts or contract standards spanning a diverse array of domains. These case studies also form the basis of our evaluation. We measure the costs of using *Quartz* in terms of the amount of code needed to express a contract compared to its Solidity and TLA⁺ equivalents, measure the execution efficiency of contracts generated by *Quartz* compared to handwritten alternatives, and present in-depth studies of applying *Quartz* and its model checking to contract development.

While there have been many efforts to ease contract development and to apply techniques from formal methods to contracts, *Quartz* is one of the only systems offering fully automated verification of functional properties, i.e., contract-specific invariants, rather than a fixed set of vulnerabilities. One important source of security vulnerabilities is interaction with external code [23]. *Quartz* is fully capable of modeling these interactions, distinguishing it from similar tools. This allows *Quartz* to flag issues underlying attacks like the widely-publicized compromise of the DAO contract [13].

The remainder of this paper will motivate, discuss, and evaluate the design of *Quartz* in more detail. Section 2 covers background material and presents the contract case studies used to motivate the design of *Quartz*'s language, presented in Section 3. *Quartz*'s model checking process is described in Section 4, and Section 5 covers Solidity generation. We present our evaluation in Section 6. Finally, Section 7 summarizes related work and Section 8 concludes.

2 BACKGROUND & REQUIREMENTS

In this section, we first summarize the execution semantics of contracts on the Ethereum blockchain. We then present a selection of case studies used to inform the design and feature set of the *Quartz* DSL. We identify common contract design patterns and motivating use cases for specific language primitives.

2.1 Ethereum Contract Execution

Contracts in Ethereum are expressed as bytecode for the Ethereum Virtual Machine (EVM), although contracts are mainly written in Solidity, a higher-level language compiled to bytecode. Ethereum's distributed ledger is maintained by a peer-to-peer network of nodes each running a local instance of the EVM. Each item added to the ledger triggers an execution of bytecode, which enacts some transformation against a smart contract. All participants stay synchronized to the latest state of the blockchain by reaching consensus on the sequence of transactions and then locally executing the corresponding bytecode. Ethereum meters bytecode execution by associating each EVM operation with a "gas cost." An upper limit on the total gas cost of each transaction (ledger item) ensures termination, and administering a transaction fee based on its total gas discourages loading the network with spurious computations.

Ethereum treats the exchange of virtual tokens, "Ether," as a first-class primitive. Every contract deployed to the ledger has an associated token balance, and these tokens are used to pay the transaction fees described above. End users may also create accounts that are recorded on the ledger and maintain a balance of tokens on their behalf but are not associated with any executable code. Both these accounts and smart contracts are uniquely identified by a 256-bit public key, known as an address. A contract transformation may involve sending tokens in its possession to an address, which could be either a simple user account or another contract. The latter case is much like a function call, in that control transfers to the receiving contract, which may execute its own code. Thus, the execution of contract code is triggered by an end user explicitly invoking a transaction against that contract or implicitly by the receipt of tokens.

2.2 Contract Case Studies

Table 1 summarizes the contracts we studied in detail to motivate and refine *Quartz*'s language design. Our case studies were drawn from Ethereum standards, subjects of prior work, and real Solidity projects. These contracts encompass a diverse range of use cases such as financial applications (ERC-20, Auction), voting and decision making (ERC-1202, DAO), and data provenance (Logistics). Several involve intricate security concerns and rich featuresets.

We identified several common and recurring design patterns in these contracts, generally in line with prior work [4, 45]:

- Contracts typically go through a multi-phase lifecycle. In each phase, a different set of transactions are permitted.
- Authorization is a primary concern for contract code. Most actions are only intended to be performed by a specific set of principals.
- Timing is also a major concern. Some actions are only valid within a specific time range.
- Actions may be conditionally enabled, depending on the contract's internal state.

In short, contract use cases typically center on maintaining internal state and restricting when, how, and by whom this state is modified, in an effort adhere to a known protocol.

Additionally, we observed frequent use of certain language features in the implementations of these contracts:

- Both signed and unsigned integer arithmetic, the latter frequently to implement blockchain-managed tokens or assets
- Use of structs to simplify management of complex state and use of maps to track possession of virtual tokens or shares

Name	Description			
Auction [21]	Simple auction with open participation and			
	open bids			
Crowdfunding [45, 50]	Crowdfunding campaign with deadline and			
	token refund logic			
Logistics	Shipment tracking contract			
SimpleMultiSig	Two-participant multi-signature wallet			
StaticMultiSig [43, 44]	Multi-signature wallet with fixed set of			
	signers			
DynamicMultiSig	Multi-signature wallet with dynamic set of			
	signers			
ERC-20 [43, 58]	Ethereum standard token implementation			
ERC-721 [16, 43]	Ethereum standard non-fungible token im-			
	plementation			
ERC-1202-Simple [59]	Ethereum standard voting implementation			
ERC-1202-Weighted [59]	Voting implementation with voter-specific			
	weights			
ERC-1540 [25, 30]	Ethereum standard asset management im-			
	plementation			
ERC-1630 [8, 27]	Ethereum standard time-based fund distri-			
	bution implementation			
ERC-1850 [7]	Ethereum standard token loan implemen-			
	tation			
ERC-780 [55]	Ethereum standard metadata registry			
RockPaperScissors [14]	Simple rock-paper-scissors game imple-			
	mentation			
DAO [51]	Decentralized autonomous organization			

Table 1: Contract Case Studies

 Hashing to implement cryptographic commitment schemes (e.g., in multiplayer games) or to verify possession of some secret for capability-based access control

Notably, we did not identify the need for complex branching logic. Transactions in the contracts we surveyed typically begin by asserting a number of conditions, aborting if they aren't satisfied, and otherwise executing a simple sequence of statements. We similarly did not observe frequent use of looping constructs. Finally, all of the case studies we analyzed can be implemented within a single contract. While some were implemented using a suite of cooperating contracts, this design choice was not fundamental to achieving the desired functionality.

3 LANGUAGE DESIGN

This section presents the design of the *Quartz* domain-specific language. We first describe the structure of a *Quartz* contract description and then present a running example contract that we will use throughout the paper. Next, we formalize *Quartz*'s syntax and operational semantics.

3.1 Language Structures

A contract definition in *Quartz* consists of an extended finite-state machine and an optional set of properties to verify about the machine's behavior. A state machine in turn consists of three pieces. The first is an optional sequence of definitions of any Struct types to be used within the contract. The second piece is a set of *fields*, each given a unique name and annotated with a type. *Quartz* supports simple types such as Int and Uint, parameterized types such as Maps and Sequences, and Struct types. *Quartz* also includes

types specifically useful for contract development such as Identity (a unique identifier for a ledger participant) and a Timespan type. A HashValue type is parameterized by a sequence of types indicating the structure of its preimage. It only supports equality checks with instances of the same type. This encourages the use of hashing for purposes like commitment schemes and capability-based access control while discouraging the use of hashing as a pseudo-random number generator, a practice that has introduced vulnerabilities in past contracts [3].

The last component of a state machine is a set of state *transitions*. Each transition is comprised of the following elements:

- A unique name, used to invoke the transition
- A source state and destination state
- A set of parameters, each given a name and type
- A guard, written as a predicate over the machine's fields and the transition's parameters
- An *authorization predicate* restricts which parties may trigger a transition
- A *body*, written as a sequence of statements executed for their side effects

Quartz state machines are event triggered, and a transition is only eligible for execution if its guard is satisfied. The statements within a transition body are kept simple to facilitate model checking, with no branching constructs. They may either modify a field or transfer tokens to an external contract. An authorization predicate determines who may initiate execution of the associated transition, a particularly important concern for smart contract applications. Quartz's authorization clauses allow contract authors to express rich semantics that are cumbersome to express using guards alone. They are built from three terms of the form i, satisfied when Identity iapproves the transition, and the forms any(I) and all(I), where Iis of type Sequence[Identity]. These are satisfied when one or all members of the referenced group approve, respectively. These terms may be arbitrarily combined with Boolean && and || operators.

The second, optional element of a contract description is a set of *invariants* regarding the state machine's possible execution traces. These are written as predicates over the state machine's fields, with some additional primitives. Predicates may refer to transition parameter values or to an aggregate sum over a Sequence or Mapping type. Additionally, a predicate can use min or max to refer to the minimum or maximum value that a variable assumes over the lifetime of the state machine. For example, given a state machine containing transition t with parameter p, max(t.p) refers to the maximum value of p ever used in an execution of t. This allows *Quartz* to check rudimentary temporal properties [41, 49].

3.2 Running Example

We will use the running example of a smart contract for administering a simple auction on a blockchain. The purpose of this contract is to accept a sequence of ascending bids, each backed by a token deposit, from any potential party for an item put up by a specific seller, who is responsible for deploying the contract. It is up to the contract's implementation code, which is executed exactly as written by the underlying distributed ledger, to properly enforce the auction's terms. First, the issuer of the highest bid, regardless of their identity, is duly recorded as the winner. Bids are accepted up



Figure 2: A State Machine for an Auction Contract

10

11

12

13

15

16

17

18

19

20

21

22

23

24

25

26

27 28

29

30

31

32 33

34 }

until a publicly declared deadline, which cannot be adjusted after the start of the auction by any party. Any principal who issued a losing bid is able to recover their tokens, while the seller may only claim the auction's proceeds once it is closed.

Our auction features four phases of operation. Its representation as a state machine is shown in Figure 2. Each transition is annotated with a guard, written above the horizontal line, and actions shown below the horizontal line. When the contract is deployed, its creator is recorded as the seller. The contract begins its life in the **Init** phase, awaiting the first bid. As soon as a bid arrives, the bid and its sender are recorded, and the contract transitions to the **Open** phase. Here, an arbitrary number of subsequent bids may be received and recorded, as long as each exceeds the previous highest bid. Unlike in the previous phase, the contract now must also refund the newlysupplanted highest bidder. Finally, once the auction's deadline has passed, any party may move to close the auction. Only once the **Closed** phase is reached can the seller claim their earnings, with a transition into the **Redeemed** phase.

3.3 Language Syntax

Figure 3 gives the *Quartz* implementation of the auction state machine shown above. A formal presentation of *Quartz*'s syntax is also given in Appendix A. *Quartz* supports the standard arithmetic and Boolean operators, comparisons, and both in and not in operators to check for membership in an object of Sequence type. Literals of type Bool, Int, Uint, and Timespan are written as expected, with the possible exception of a Timespan instance, written as an integer followed by a unit such as minutes or hours.

Quartz transitions begin with a header of the form *source* -> (*Parameters*) *destination*. This is followed by an optional requires block to express a guard and an optional authorized block to express an authorization predicate. Finally, the body of the transition is enclosed within braces and consists of a sequence of simple statements, like assignment to a field.

The *Quartz* language contains several contract-specific features. Many distributed ledgers, most notably Ethereum, have first-class support for virtual currency that may be bound to contracts and exchanged among them. State machines in *Quartz* use keywords to check their balance or disburse tokens to an external contract. If we wish to produce contracts for a ledger without first-class tokens, we can emulate this functionality by adding an extra field and the

```
contract Auction {
    data {
        Seller: Identity
        HighestBid: Uint
        HighestBidder: Identity
        Duration: Timespan
        Deadline: Timestamp
    3
    initialize: ->(duration: Timespan) init {
        Seller = sender
        Duration = duration
        HighestBid = 0
    }
    initialBid: init ->(tokens: Uint) open {
        Deadline = now + Duration
        HighestBid = tokens
        HighestBidder = sender
    }
    submitBid: open ->(tokens: Uint) open
    requires [ tokens > HighestBid && now <= Deadline ] {</pre>
        send HighestBid to HigestBidder
        HighestBid = tokens
        HighestBidder = sender
    3
    close: open -> closed requires [ now > Deadline ]
    redeem: closed -> redeemed authorized [ Seller ] {
        send HighestBid to Seller
```

Figure 3: An Auction Contract Written in Quartz

necessary operations to the generated implementations. Transition authorization is treated as a first-class primitive in *Quartz*, unlike in Solidity and other contract languages. *Quartz* allows contract authors to express rich authorization constraints such as restricting an operation to any member of a particular group or requiring approval from all members of a group before it is executed. Finally, *Quartz* restricts communication between state machines. A state machine may send tokens to another state machine, but it cannot invoke another machine's transitions directly. This simplifies the expression and verification of contract logic. Note that *Quartz* makes no assumptions about the behavior of the recipient, which may or may not be another *Quartz* state machine, for model checking.

3.4 Language Semantics

Here, we formally define a subset of the operational semantics of the *Quartz* DSL. Evaluation rules for expressions, which are generally routine, are omitted for brevity. A *Quartz state machine* is formally defined as a 4-tuple $\langle Q, q_0, T, F \rangle$ where Q is a set of states, q_0 is the initial state, T is a set of *transitions*, and F is a set of fields, each with a specific name and type. A transition $t \in T$ is defined as a 7-tuple $\langle name, src, dst, g, a, P, B \rangle$ where:

- *name* is the transition's unique name.
- src and dst are the transition's source and destination states.
- *g* is a boolean-valued *guard* expression.
- *a* is an *authorization* predicate.
- *P* is a set of transition-specific parameter values, each with a name and type.
- *B* is the transition's *body*: a sequence of statements executed when the transition fires.

A *Quartz* state machine is event triggered. An external, addressable entity (either an end user or another contract) invokes a transition by submitting a message $m = \langle i, t, V \rangle$ to the state machine specifying its identity *i*, a transition *t* to execute, and a (possibly empty) set *V* of values for *t*'s parameters.

The current status *S* of a state machine is defined as a 7-tuple $\langle q, \sigma, \alpha, M, s, \gamma, C \rangle$ where:

- *q* is the machine's current state
- σ is a mapping from names to values representing the current *context*. It always includes the state machine's fields as well as built-in values **sender**, **balance**, and **now**.
- *α* is a mapping from transitions and transition parameter values to the set of identities that have authorized invocation of that transition with that particular set of parameter assignments.
- *M* is a queue of transition invocation messages. The machine may only read the head of the queue or remove the head of the queue. It may not inspect any other queue elements or the queue's length.
- *s* indicates which statement within a transition to execute. The special statement form \top indicates that the machine has not yet started execution of any transition, while \perp indicates that a transition has just completed. Thus when executing some transition with body $B = [b_1, \ldots, b_n]$, *s* will assume the sequence of values $[\top, b_1, \ldots, b_n, \bot]$.
- γ is an authorization clause that must be evaluated to determine if the invoked transition may proceed, or □ if no such determination is in progress.
- *C* is a stack of transitions currently in progress.

A machine's initial status, before it executes its initial transition, is therefore $\langle q_0, \{f.name \mapsto 0_{f.type} : f \in F\}, \emptyset, \varepsilon, \top, \Box, \varepsilon \rangle$, where ε denotes an empty sequence and 0_T denotes the zero element of type *T*.

3.4.1 Transition Authorization. A subset of *Quartz*'s operational semantics for transition authorization is presented in Figure 4. In all rules, \rightarrow indicates small-step evaluation and \Downarrow indicates big-step evaluation. \Downarrow_{σ} more specifically denotes big-step evaluation with σ as the initial environment. The \circ symbol denotes concatenation, e.g., $m \circ M$ signifies the element m prepended to the sequence M.

When a new message *m* is dequeued, the machine's status is updated to reflect that the message's sender, *i*, approves of the transition's execution, and *Quartz* begins evaluation of the relevant transition's authorization clause, *t.a*, as shown in UPDATEAUTH. Terms in the authorization clause are of the forms *i*, any(I), and all(I). These terms are evaluated as expected, shown in AUTHSIN-GLETRUE, AUTHANYTRUE, and AUTHALLTRUE. Each of these rules has a complement, e.g., AUTHSINGLEFALSE, omitted for brevity. Authorization clauses may also use Boolean && and || as connectives, with the expected evaluation rules.

If we have $t.a \downarrow true$, then evaluation proceeds to the transition's body. As shown in AUTHSUCCESS, the current statement in the machine's status advances from \top to b_1 , the first element of t.b. Otherwise, *Quartz* discards message m, removes the transition's parameter assignments from the current context σ , and awaits the next incoming message. However, the sender's approval of the transition persists in α and affects future evaluations of t.a.

$S = \langle q, \sigma, \alpha, m \circ M, \top, \Box, C \rangle \qquad m = \langle i, t, V \rangle$
$q = t. \operatorname{src} \sigma' = \sigma \cup V$
$t.g \downarrow_{\sigma'} true \qquad \alpha_{t,V} = \alpha[\langle t, V \rangle]$
$S \to \langle q, \sigma', [\langle t, V \rangle \mapsto \alpha_{t,V} \cup \{i\}] \alpha, m \circ M, \top, t.a, C \rangle $ (UpdateAuth)
$S = \langle q, \sigma, \alpha, m \circ M, \top, i, C \rangle$ $\frac{m = \langle j, t, V \rangle \qquad i \in \alpha[\langle t, V \rangle]}{S \to \langle q, \sigma, \alpha, m \circ M, \top, \text{ true, } C \rangle} $ (AuthSingleTrue)
$ \begin{array}{c} S\langle q,\sigma,\alpha,m\circ M,\top,\operatorname{any}(I),C\rangle\\ \underline{m=\langle j,t,V\rangle \existsi\in I:i\in\alpha[\langle t,V\rangle]}\\ \overline{S\to\langle q,\sigma,\alpha,m\circ M,\top,\operatorname{true},C\rangle} \end{array} (\operatorname{AuthAnyTrue}) \end{array} $
$ \begin{split} S &= \langle q, \sigma, \alpha, m \circ M, \top, \mathrm{all}(I), C \rangle \\ \underline{m &= \langle j, t, V \rangle \qquad \forall i \in I : i \in \alpha[\langle t, V \rangle]}_{S \to \langle q, \sigma, \alpha, m \circ M, \top, \mathrm{true}, C \rangle} (\mathrm{AuthAllTrue}) \end{split} $
$ \begin{array}{c} S = \langle q, \ \sigma, \ \alpha, \ m \circ M, \ \top, \ \text{true}, \ C \rangle \\ \hline m = \langle i, \ t, \ V \rangle & t.B = [b_1, \ \dots, \ b_n] \\ \hline S \to \langle q, \ \sigma, \ \alpha, \ m \circ M, \ b_1, \ \Box, \ C \rangle \end{array} (\text{AuthSuccess}) $
$S = \langle q, \sigma, \alpha, m \circ M, \top, \text{ false, } C \rangle$ $\underline{m = \langle i, t, V \rangle \sigma' = \sigma - V}$ $\overline{S \to \langle q, \sigma', \alpha, M, \top, \Box, C \rangle} \text{(AuthFailure)}$

Figure 4: Quartz Operational Semantics: Authorization

3.4.2 Transition Execution. A transition's body consists of a sequence of statements, evaluated in order when the transition executes. Evaluation rules for *Quartz*'s small set of statement types are presented in Figure 5 and are relatively straightforward. When the end of the transition's body, \perp , is reached, *Quartz* consults the state machine's stack for a record of an in-progress parent transition, consisting of its environment σ and the next statement to execute s. If such a record exists, control returns to the parent transition. Otherwise, the state machine moves to the implicit \top statement. In both cases, the message at the head of the queue *M* is finally removed, the current environment σ is stripped of the transition's parameter assignments, and prior approvals of the transition's execution in α are cleared.

Quartz's send statement has slightly more complex evaluation rules. Because a send involves ceding control to the recipient, its

$$\frac{s = "x = v" \qquad x \Downarrow_{\sigma} x' \qquad v \Downarrow_{\sigma} v'}{S \to \langle q, [x' \mapsto v']\sigma, \alpha, M, \operatorname{next}(s), \Box, C \rangle}$$
(EVALASSIGN)

$$\frac{S = \langle q, \sigma, \alpha, M, s, \Box, C \rangle}{v \Downarrow_{\sigma} v' \times \downarrow_{\sigma} x'} \frac{s = \text{``append } v \text{ to } x''}{\sigma[x'] = [x_1, \dots, x_n]} \xrightarrow{\text{(EVALAPPEND)}}$$

$$\frac{S \to \langle q, [x' \mapsto [x_1, \dots, x_n, v'] \sigma, \alpha, M, \text{next}(s) \Box, C \rangle}{s \to (x_1, \dots, x_n, v'] \sigma, \alpha, M, \text{next}(s) \Box, C \rangle} \xrightarrow{\text{(EVALAPPEND)}}$$

$$S = \langle q, \sigma, \alpha, M, s, \Box, C \rangle$$

$$s = \text{``clear } x^{\text{``}} \times \bigcup_{\sigma} x^{\text{'}}$$

$$S \to \langle q, [x^{\prime} \mapsto \varepsilon]\sigma, \alpha, M, \text{next}(s), \Box, C \rangle$$
(EVALCLEAR)

$$S = \langle q, \sigma, \alpha, m' \circ m \circ M, \bot, \Box, c \circ C \rangle$$

$$c = \langle \sigma', s \rangle \qquad m' = \langle i, t, V \rangle$$

$$S \to \langle t.dst, \sigma', [\langle t, V \rangle \mapsto \emptyset] \alpha, m \circ M, s, \Box, C \rangle$$
(FINISHTRANSITION1)

 $S = \langle q, \sigma, \alpha, m \circ M, \bot, \Box, \varepsilon \rangle \qquad m = \langle i, t, V \rangle$ $S \to \langle t.dst, \sigma, [\langle t, V \rangle \mapsto \emptyset] \alpha, M, \top, \Box, \varepsilon \rangle$ (FinishTransition2)

where

$$t.b = [b_1, b_2, \dots, b_n]$$
$$next(b_i) = \begin{cases} b_{i+1} & \text{if } i < n \\ \bot & \text{otherwise} \end{cases}$$

Figure 5: Quartz Operational Semantics: Statements

evaluation non-deterministically produces one of several possible outcomes, each shown in Figure 6. In the first case, SENDSUCCESS, the send completes without any disruption to control flow. Execution proceeds to the next statement. Otherwise, the recipient may either throw an exception, halting progress as denoted in SENDER-ROR, or it may reenter the state machine by invoking an arbitrary transition, expressed in SENDREENTER. This final evaluation rule is the only means by which the machine's stack *C* may grow.

$$\begin{split} S &= \langle q, \sigma, \alpha, M, s, \Box, C \rangle \qquad s = \text{``send } a \text{ to } i^{``} \\ a \Downarrow_{\sigma} a' \qquad a' \leq \sigma [\text{balance}] \qquad b = \sigma [\text{balance}] - a' \\ \hline S \to \langle q, \text{ [balance} \mapsto b] \sigma, \alpha, M, \text{ next}(s), \Box, C \rangle \qquad \text{(SendSuccess)} \\ \hline \\ \hline S \to \langle q, \sigma, \alpha, M, s, \Box, C \rangle \qquad s = \text{``send } a \text{ to } i^{``} \\ \hline S \to \text{ERROR} \qquad \text{(SendError)} \\ \hline \\ \hline \\ S &= \langle q, \sigma, \alpha, m \circ M, s, \Box, C \rangle \qquad m = \langle i, t, V \rangle \\ s &= \text{``send } a \text{ to } j^{``} \qquad a \Downarrow_{\sigma} a' \qquad j \Downarrow_{\sigma} j' \\ \sigma' &= [\text{balance} \mapsto \text{balance} - a'] \sigma \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ \hline \\ S &\to \langle q, \sigma' - V, \alpha, m' \circ m \circ M, \top, \Box, c \circ C \rangle \qquad \text{(SendReenter)} \end{split}$$

Figure 6: Quartz Operational Semantics: send

3.5 Semantic Analysis

The *Quartz* compiler performs a variety of checks on a contract description before it attempts to generate TLA⁺ or Solidity code. A well-formed state machine has a unique initial transition with no guard or authorization predicate. Any state mentioned in the description must be reachable from the initial state following some sequence of naive transitions (i.e., neglecting guards and authorization predicates). *Quartz* ensures that no parameter shadows a field due to a name collision and that all guards, authorization predicates, and statements refer to well-defined variables. *Quartz*

also performs simple type checking to avoid errors in subsequent code generation phases. Much of this is as expected, for example checking that arithmetic and logical operators are only performed to compatible types. Guards must be Boolean-valued, the target of the in and not in operators must be a Sequence, token sends only target expressions of Identity type, and so on.

4 CONTRACT VERIFICATION

This section describes how *Quartz* verifies contracts. *Quartz* translates a contract definition to a specification expressed in TLA⁺ [34]. The automatically generated specification captures both the contract state machine's logic as well as the semantics of its execution environment, namely an Ethereum-based distributed ledger. *Quartz* feeds this specification into TLC, an explicit-state model checker for TLA⁺ that enumerates and searches execution traces for violations of user-provided properties, in this case the invariants written by the contract author in her description. This approach has the advantage of being fully automated, with no intervention needed from the contract author, but does raise the challenge of bounding the execution search space so that model checking terminates.

4.1 Why Model Checking and TLA+?

We chose bounded model checking as *Quartz*'s core verification technique because it does not require significant intervention from the end user, i.e., the contract author. Although a contract author must write the invariants she would like to have verified, *Quartz* fully automates the more difficult task of writing a formal specification of the contract's behavior and its execution environment that is suitable as input to a model checker. Model checking also offers immediately useful feedback to the user as output — an execution trace that produces a violation of one or more of the desired properties. This feedback helps guide a contract author in making refinements to her state machine.

TLA⁺ serves as *Quartz*'s target specification language and its verification backend. TLA⁺ and its model checker, TLC, are relatively mature, well-documented, and have been successfully applied in developing and testing significant systems [40]. More modern model checkers have since emerged, but they tend to be inherently tied to the semantics of particular implementation languages such as C [26] or operate at the low level of bytecode [24]. The flexibility of TLA⁺'s specification language simplifies *Quartz*'s task of generating a formal contract specification. This becomes especially important when describing the execution semantics of Solidity, which have important differences from the semantics of traditional programming languages. Moreover, there are ongoing efforts to modernize verification in TLA⁺, such as symbolic model checking with SMT solvers [32], that *Quartz* may be able to use in the future.

4.2 Specification Generation

Quartz's specification generator targets PlusCal, an intermediate language built on top of the original TLA⁺ specification language. PlusCal for pieces of the Auction case study is provided in Appendix B. PlusCal offers several features that make it a more natural target than TLA⁺ itself, such as procedures to model state transitions and conditionals to model transition guards. Translating Quartz data types and transitions into PlusCal is straightforward, but modeling *Quartz*'s authorization semantics and the blockchain execution environment is more challenging. The PlusCal generated by *Quartz* is translated into TLA⁺ with off-the-shelf tools.

Data Types. Every data type in Quartz maps to a counterpart in PlusCal. Many have direct equivalents such as Ints and Maps. Quartz defines the domain of the Identity type as a fixed set of symbolic constants, with a user-configurable size. The translation of a HashValue is more subtle. Quartz has no need to model hash functions in detail aside from the fact that they are assumed to be injective, nor does it require that HashValue instances are ordered. The output of hash (x_1, \ldots, x_n) is simply modeled as a PlusCal tuple (x_1, \ldots, x_n) , preserving injectivity and enabling equality checks among HashValue instances.

Transitions. Each transition defined in a *Quartz* contract's state machine is generated as a PlusCal procedure, with transition parameters naturally mapping to procedure parameters. An extra parameter is added to the PlusCal procedure to track the transition's sender. An auxiliary field is used to track the machine's current state. The procedure body begins with three conditional checks: one to ensure that the state machine is currently in the transition's designated starting state, a second to ensure that the transition's authorization predicate, if defined, is also satisfied. Finally, the statements forming the transition body are converted to PlusCal in the expected way, as PlusCal supports a standard collection of arithmetic and logical operators.

Authorization. Quartz adds auxiliary fields to a contract's Plus-Cal specification to accurately model the authorization semantics detailed in Figure 4. Note that an entity approves of the execution of a transition for a particular set of parameter values. For example, consider a transition T with input parameters of types t_1, \ldots, t_n that includes a term of the form all(I) in its authorization predicate. Quartz generates a PlusCal function (associative array) F of type $t_1 \times \cdots \times t_n \times \text{Identity} \to \text{Boolean. Then, all}(I)$ is evaluated in PlusCal, which has native support for quantifiers, as $\forall i \in I : F(p_1, \ldots, p_n, i)$, where p_i is the i^{th} transition argument. Similar translations are performed as needed for the other authorization term forms. Quartz generates the minimum number of auxiliary fields, only when authorization predicates cannot be satisfied by just a single transition approval.

Modeling the Environment. Once an Ethereum contract is deployed to a blockchain, any of its transformations may be invoked at any time, by any user. For a *Quartz* contract, this means any of the state machine's transitions may be invoked at any time. The PlusCal model generated by *Quartz* is organized around a main invocation loop that simulates this environment. Each time through the loop, a transition t is non-deterministically selected for execution. Values for its input parameters v_1, \ldots, v_n are similarly selected from their respective domains, including an identity i as sender.

The second major challenge in modeling the Ethereum execution environment is capturing the behavior of sending tokens from one contract to another, i.e., the semantics of Figure 6. As explained above, there are two primary means of exchanging tokens between one Ethereum contract and another: using Solidity's transfer primitive or using the call primitive. Both yield control to the destination contract. call is more flexible in that it allows the recipient to execute arbitrary code, but this may include a reentrant invocation of the sending contract. transfer restricts execution but propagates any exceptions thrown by the receiver back to the sender, which may block forward progress.

The user may specify the use of transfer or call as a configuration parameter. *Quartz* is capable of modeling either primitive's behavior. The generated PlusCal model deducts from the balance field and then makes a non-deterministic choice to model the recipient's response. When modeling transfer, the recipient either does nothing, indicating a routine token transfer, or throws an exception. When modeling call, the recipient either does nothing, meaning any code executed by the recipient had no consequence for the sender, or it non-deterministically selects some transition t of the sender's to invoke, modeling possible re-entrant execution.

The behavior of Ethereum's exceptions cannot be expressed in PlusCal. Instead, *Quartz* generates an initial PlusCal specification, invokes the PlusCal translator to produce TLA⁺, and modifies this code directly. The final TLA⁺ generated by *Quartz* for model checking formalizes unwinding of the stack upon an exception: reverting the contract's fields to their state before the current call chain and jumping to the main invocation loop to begin a fresh transition.

4.3 Bounding the Search Space

The TLA⁺ specifications generated by *Quartz*, as described so far, have an infinite execution space. We must apply bounds to this search space to ensure that model checking terminates. *Quartz* exposes a set of parameters that the user may set at verification time. All of these parameters convey aspects of the contract's execution or the domain of data types:

- Minimum and maximum integer values
- Number of distinct Identity instances to model
- The maximum call depth reached during transition execution
- The maximum number of iterations of the main transition invocation loop

The model checker is essentially exploring all possible sequences of state machine transitions. The first two parameters above limit the branching factor of the search space, while the third and fourth limit the depth to which it is explored.

5 SOLIDITY GENERATION

Quartz is able to translate state machine descriptions to Solidity implementations, enabling seamless deployment once a contract has been sufficiently validated by its developer. *Quartz* targets Solidity rather than EVM bytecode for several reasons. Solidity is more human readable than EVM bytecode, which means a contract author may easily inspect and audit a generated implementation if necessary. Moreover, there is an ongoing effort within the Ethereum community to replace the original EVM with a new virtual machine based on WebAssembly [20]. By targeting Solidity, *Quartz* remains agnostic to this potential change.

Data Types. The data types for state machine fields and transition parameters in *Quartz* each have a natural analogue in Solidity. For example, an Identity corresponds to a Solidity address. More recent versions of Solidity require address variables that are the target of a send to be explicitly annotated with the payable keyword. When *Quartz* generates Solidity, it infers all necessary uses of the payable designation.

State Transitions. A state transition maps to a Solidity function, where transition parameters have the expected correspondence to function parameters. An auxiliary field represents the *Quartz* machine's current state, and a requires statement at the beginning of the function ensures that the machine is in the proper starting state, otherwise the function (transition) does not proceed. If the transition defines a guard, it is evaluated within a second requires statement. Translation of the transition's body is straightforward, as Solidity features all of the usual operators and assignment semantics. Finally, the machine's current state is updated to the designated destination state of the transition.

Transition Authorization. As with PlusCal, Quartz augments a Solidity contract with additional fields for tracking prior authorizations when needed, modifies these fields when a transition with an authorization predicate is invoked, and checks these fields within a conditional to ensure that the authorization predicate is satisfied before a transition is executed. Unlike for guards, authorization predicates are not checked within a requires as this would revert the action of recording the sender's approval for the transition. Solidity does not have the same flexibility as TLA⁺, and generating fields to track prior authorization for each possible combination of input parameters. Given a transition with parameters of type $p1, \ldots, pn$, Quartz generates Solidity that hashes the concatenation of the parameter values and uses the result to look up prior authorizations.

Quartz-generated Solidity uses mapping instances to efficiently record and look up prior transition authorizations. However, groups of entities referred to by a Quartz any or all authorization term are represented as arrays, which means we must use loops for certain operations on these groups. This is because Solidity mappings do not permit iteration over their members and therefore cannot be used to emulate something like a set data structure. In addition, the Quartz generation logic must produce code that is as flexible as possible, with no assumptions or domain knowledge about how identity groups are used or modified over time. A Quartz Sequence of Identity values could be modified by a clear statement at any time, and there is no efficient way to empty a Solidity mapping.

Loops can be problematic in Solidity because of termination and gas cost issues. However, loops generated by *Quartz* are only used for simple scans of finite data structures, and as we show in our evaluation below, simple *Quartz*-generated Solidity with loops often actually has lower gas costs for practical workloads than handwritten code that uses more complex data structures to avoid them. If warranted, we could extend *Quartz* to generate code following Solidity's iterable mapping pattern¹ or to perform additional optimizations when an identity group is never cleared.

6 EVALUATION

This section presents our evaluation of *Quartz*. We begin by measuring the amount of code required to express various contracts in *Quartz* versus in Solidity and the amount of code generated from a *Quartz* description. Next, we discuss results from the model checking process in depth for two case studies. Finally, we compare the execution costs of *Quartz*-generated and handwritten contracts.

6.1 Contract Size

We use lines of code as a proxy for code complexity and required developer effort. Here, we are interested in the extent to which *Quartz* enables concise expression of rich contract logic, particularly when compared to Solidity, the de facto standard programming language for smart contracts. We also seek to determine if *Quartz* introduces overhead by generating significantly more verbose Solidity code than would be produced by a Solidity programmer. This is particularly relevant in the blockchain setting, where the size of a contract's compiled bytecode directly impacts the gas costs of deploying the contract to the ledger. Finally, we quantify developer effort saved when using *Quartz* for model checking by measuring the size of a contract's TLA⁺ representation.

Table 2 shows the lines of code needed to express all of our case studies in *Quartz* and in Solidity. It also shows the lines of Solidity and TLA⁺ generated from the *Quartz* implementation of each contract. Each row in the table corresponds to one of the case studies presented in Table 1. When writing Solidity for each case study, we modified an existing code base (cited in Table 1) whenever possible rather than starting from scratch to ensure we produced idiomatic Solidity code. We were also careful to remove any comments or extraneous lines of code, like redundant getter functions that have no *Quartz* equivalent, to make the comparison as fair as possible.

As shown in the table, every case study's implementation in *Quartz* involved fewer lines of code than the handwritten Solidity equivalent, with the exception of the DAO, where the two are equal. The ratio of lines of *Quartz* to lines of handwritten Solidity is 0.68 on average and as small as 0.34, for the StaticMultiSig case sutdy. *Quartz* was particularly concise for the multi-sig wallets and RockPaperScissors because of its state machine structure and authorization predicates. These allow a *Quartz* developer to avoid writing tedious and repetitive assertions at the beginning of contract transactions to verify that the contract is in the expected state before proceeding.

Quartz typically, but not always, generates Solidity code that is more verbose than the handwritten equivalent. There are cases where the generated code is shorter than the handwritten code. This is often the case when the handwritten Solidity leverages domain knowledge to use more verbose but arguably more efficient data structures, particularly to track authorization, than the general code produced by *Quartz*.

Finally, significantly more TLA⁺ code is needed to express each contract than Solidity or *Quartz* code. This is mainly because a contract's TLA⁺ specification expresses both the contract's logic and its execution semantics. In particular, the TLA⁺ representation of any contract must describe the main invocation loop in which any user may invoke any of the contract's transitions with arbitrary parameter values. It must also express the potential for reentrant execution after a send and exception handling. Therefore, even a short contract generates a relatively lengthy TLA⁺ specification.

¹https://github.com/ethereum/dapp-bin/blob/master/library/iterable_mapping.sol

Case Study	Quartz	Handwrt.	Gen.	TLA ⁺
		Solidity	Solidity	
Auction	33	40	53	205
Crowdfunding	31	39	50	193
RockPaperScissors	39	76	75	261
Shipping	20	35	46	172
SimpleMultiSig	12	34	30	137
StaticMultiSig	13	38	44	125
DynamicMultiSig	15	39	50	141
ERC-20	43	49	50	209
ERC-721	48	59	86	227
ERC-1202-Simple	24	50	49	197
ERC-1202-Weighted	42	57	73	261
ERC-1540	114	143	181	464
ERC-1630	22	23	37	161
ERC-1850	110	182	242	588
ERC-780	16	17	24	140
DAO	126	126	150	390

Table 2: Lines of Code to Express Case Studies

6.2 Model Checking

Here, we describe experiences model checking two contracts using *Quartz*. For both contracts, *Quartz* helps to surface non-obvious bugs that could easily be overlooked during contract development. Below, we report the time required for model checking to find invariant violations. These times were obtained on a workstation with an Intel i7-6700 CPU and 32 GiB of RAM running version 2.13 of the TLC model checker with 8 worker threads.

6.2.1 Model Checking an Auction Implementation. We introduced the Auction case study in detail in Section 3.2. Its state machine appears in Figure 2 and *Quartz* code appears in Figure 3. While it may appear perfectly logical, the contract as presented above features multiple security vulnerabilities, related to it distribution of repayments back to surpassed bidders and to the seller. These vulnerabilities are particularly insidious because they emerge from code that appears innocuous. They are good examples of how Ethereum's execution semantics differ from those of traditional software and can trip up contract authors.

To begin, consider the following invariant for the auction:

 $p_1 : \mathbf{Closed} \Rightarrow \mathsf{HighestBid} \ge \max(\mathsf{submitBid}.\mathsf{tokens})$

This property takes advantage of a number of *Quartz*'s features for writing invariants. It states that if the auction reaches the **Closed** state, then the value of HighestBid should be greater than or equal to the maximum value ever assigned to the tokens parameter for the submitBid transition.

Recall that *Quartz* may generate a contract that uses either Ethereum's transfer construct or the call construct for dispensing currency. The choice is configurable by the user, and *Quartz* is fully capable of generating TLA⁺ to model either. If transfer is used, the model checker finds the following violation of p_1 . This required only 2 seconds to complete on our test system.

- (1) Identity I_1 deploys a new auction contract. The auction enters the **Init** state.
- (2) I_2 submits an initial bid of 2 tokens and is recorded as the highest bidder. The auction enters the **Open** state.

- (3) I_3 submits a new bid of 4 tokens. The auction sends 2 tokens back to I_2 as a refund, since they are no longer the highest bidder.
- (4) I₂ reacts by throwing an exception. This propagates back to the auction contract (due to the use of transfer) and the current transition is aborted. I₃'s bid is lost.
- (5) No additional bids are submitted before I_1 moves to close the auction and I_2 is declared the winner.

Here, I_2 is able to hijack the auction and prevent itself from being supplanted as the highest bidder, rigging the results of the auction.

 p_1 is also violated if we use call rather than transfer, again because of an issue in refunding a previous bidder. TLC found the following trace in 6 seconds on our test system.

- (1) Identity I_1 deploys a new auction contract. The auction enters the **Init** state.
- (2) *I*₂ submits an initial bid of 2 tokens and is recorded as the highest bidder. The auction enters the **Open** state.
- (3) I_3 submits a new bid of 3 tokens. The auction sends 2 tokens to I_2 as a refund.
- (4) I₂ responds to the receipt of tokens by submitting a new bid of its own, with a value of 4 tokens, creating a reentrant invocation of the *submitBid* transition.
- (5) The auction accepts I₂'s bid, sets HighestBidder to I₂ and HighestBid to 4.
- (6) Control returns to the parent transition, which has just completed its send. It updates HighestBidder to I_3 and HighestBid to 3, accounting for I_3 's bid but overwriting I_2 's bid.
- (7) No further bids arrive. The auction reaches the Closed state.

Here, we see that contract re-entrancy, an issue best known for its exploitation by malicious actors, can also lead to undesirable outcomes for well-intentioned actors. One could easily imagine a developer seeking to create a contract that submits a bid on her behalf in reaction to having just been displaced as an auction's winner, possibly to implement some bidding strategy, yet that would go awry in this implementation.

To address either of these bugs, the contract author could instead store a *Quartz* Map[Identity, Uint] tracking pending refunds that is updated when a newly winning bid is submitted. A previous bidder must invoke a separate transition to ask the contract to send her a refund, decoupling this from bidding. This is a wellknown design pattern in Solidity [17], although it is prone to its own re-entrancy issues, which *Quartz* can also identify through its verification. Say we modify *submitBid* accordingly and add the following transition to allow bidders to claim refunds once the seller has redeemed their winnings:

```
refund: redeemed -> redeemed
requires [ Balances[sender] > 0 ] {
    send Balances[sender] to sender
    Balances[sender] = 0
}
```

Consider the following new invariant for the auction:

```
p_2: balance \geq 0
```

John Kolb, John B. Yang, Randy H. Katz, and David E. Culler

This states that the contract's balance cannot go negative, i.e., it cannot dispense more tokens than it receives. While this is impossible for a contract running on the Ethereum blockchain, negative contract balances are within the search space defined by *Quartz* for model checking because they can usefully indicate a contract's vulnerability to unbounded token withdrawals. Indeed, if *Quartz* is configured to model behavior of token sends using call, model checking finds the following violation in 15 seconds:

- (1) Identity I_1 deploys a new auction contract.
- (2) *I*₂ submits an initial bid of 1 token. The auction enters the **Open** state with balance = 1.
- (3) I_3 submits a new bid of 4 tokens, hence balance = 5.
- (4) No subsequent bids are submitted before the deadline, and *I*₁ moves to close the auction. The auction enters the **Closed** state.
- (5) I₁ invokes the *redeem* transition, receiving its winnings. Now, balance = 1 and the auction enters the **Redeemed** state.
- (6) I₂ invokes the *refund* transition and is sent the 1 token recorded in Balances[I₂]. Now, balance = 0.
- (7) In reaction to this receipt of tokens, *I*₂ makes a reentrant invocation of *refund*. Balances[sender] has not yet been updated in the parent transition, so the child transition's guard is satisfied.
- (8) Another send of 1 token to I₂ is attempted, and balance = -1, violating the invariant.

This execution trace illustrates the fundamental vulnerability behind the famous compromise of the DAO contract [13]. The usual advice to Solidity developers is to set a temporary variable to the amount of tokens to send, then subtract from the appropriate contract field *before* executing a send referencing the temporary variable. *Quartz* offers an alternative sendAndConsume construct that will generate such code.

6.2.2 Model Checking ERC-1540. Quartz is useful not just for identifying subtle consequences of a contract's execution semantics, but also for identifying more routine logic errors that occur during the development process. Unlike our auction contract, which we initially developed as a litmus test for *Quartz*'s ability to surface reentrancy and exception issues, we drafted an initial implementation of ERC-1540 after *Quartz* was fairly mature, chose an invariant to verify, and used *Quartz* to refine the contract.

ERC-1540 is a proposed Ethereum standard interface for an asset management contract. Among other capabilities, it allows an owner to sell shares, issue dividends, or transfer control of the asset, all of which is tracked on the blockchain. Investors issue transactions against the contract to buy and sell shares. The *Quartz* implementation of ERC-1540 is considerably more complex than the auction seen previously. It uses five states and 16 transitions. The portion of the state machine relevant for the following discussion is shown in Figure 7.

When the contract is initialized, its creator is recognized as the asset's owner. It begins in the **Unissued** state, meaning there are no outstanding shares. If this is the case, the owner is free to transfer possession of the asset to another party, as shown in the transition at the top of Figure 7. The owner may choose to move the asset to the **Issued** state by enacting the release of a fixed number of shares. The contract features additional transitions to exchange shares not



Figure 7: Part of a State Machine for ERC-1540

shown in the figure. If any single party accumulates all outstanding shares, they are allowed to declare themselves as the new owner and convert the asset back to the **Unissued** state.

We tested our initial version of the contract by verifying the following invariant.

$$p_3: sum(Shares) = TotalShares$$
 (1)

That is, we wanted to verify that all of the asset's shares are properly conserved as they change hands.

After generating a TLA⁺ specification and running it through TLC, we were informed that many of the contract's states, Including Issued were not reachable. This was because we failed to properly initialize the contract's owner field in the state machine's initial transition. We were also able to uncover an error in our arithmetic when transferring shares.

More interestingly, we initially forgot to update the asset's shares when an entity converts it from **Issued** to **Unissued**. Initially, we simply reassigned the owner field to the transition's sender as shown in the figure, forgetting to zero out TotalShares and Shares[Sender]. This enables the following trace, identified by TLC when given a *Quartz*-generated contract spec and simplified for brevity:

- (1) *I*₁ deploys a new ERC-1540 contract and is recorded as the owner.
- (2) I₁ issues 2 shares for the asset, initially owning both of them. The asset enters the **Issued** state, and we have TotalShares
 = 2 and Shares[I₁] = 2.
- (3) *I*₁ converts the asset back **Unissued**.
- (4) I_1 transfers ownership to I_2 .
- (5) I₂ decides to issue 3 shares for the asset, all initially assigned to itself. In the transition body, we set Shares[I₂] = 3 and TotalShares = 3. However, we still have Shares[I₁] = 2. Thus, sum(Shares) ≠ TotalShares.

The solution is to add two lines to the transition from **Issued** to **Unissued**, Shares[sender] = 0 and TotalShares = 0, to properly reflect the fact that the asset no longer has shares. Because ERC-1540 is more complex than the auction above, model checking takes longer. This trace was produced after TLC ran for 4 minutes on our test workstation, while the arithmetic error in share transfers required 27 minutes to find. Running times of this mangitude are not atypical for model checking.

6.3 Execution Overhead

Finally, we measured the execution efficiency of Solidity contracts generated from *Quartz* descriptions and those of equivalent handwritten Solidity contracts. The handwritten contracts are the same as those described in Section 6.1, meaning they are adapted from existing codebases when available and simplified if necessary, e.g., by removing extra getter functions, to form a fair comparison.

Execution of Ethereum contracts is metered by *gas*, a cost assigned to each virtual machine operation, to ensure termination and discourage unnecessarily expensive contract code. It is therefore natural to measure a contract's execution efficiency by the gas it consumes. To accomplish this, we deployed both generated and handwritten versions of all case study contracts to a small private blockchain backed by nodes hosted on Amazon EC2 virtual machines. All members of the network ran Geth version 1.8.26 and used Geth's *Clique* proof-of-authority consensus mechanism. This allowed us to configure the network to use a fixed gasPrice. As a result, the gas cost of a particular workload is deterministic and reproducible. It does not fluctuate with network load as it would in a proof-of-work Ethereum network.

We wrote a contract client script for each case study using Python's Web3 library. Each script deploys the generated and handwritten versions of Solidity code, invokes an equivalent sequence of transactions against both versions, and tallies all gas costs. For example, the script for the Auction case study initializes each contract and submits the same sequence of bids to both. The results of these measurements are shown in Figure 8. Each case study is represented along the *x* axis by a pair of bars. The number above each pair is the ratio of total gas costs for the generated Solidity code to total gas costs for the handwritten equivalent.

Gas costs for *Quartz*-generated Solidity contracts are competitive with those of handwritten contracts. While the overhead is 53% for the simple ERC-1630 contract, for more substantial contracts it never exceeds 20%. Interestingly, there are some case studies where the generated contract actually has *lower* gas costs than the handwritten equivalent. Upon further investigation, we found that this was usually due to *Quartz*'s use of fewer, simpler data structures in its generated code. This typically gave the generated contract a cheaper constructor and, for some case studies, cheaper transactions when operating on a smaller body of state.

The multi-signature wallets are a good example. The handwritten wallets are based on a design used in production by Parity [44] and OpenZeppelin [43] where approvals by designated signers are tracked with both a Solidity mapping instance, to emulate a set and thus enable fast membership checks, and a Solidity array to allow iteration over all signers. This design avoids loops in the critical path but also requires bookkeeping to manage both the mapping and array. *Quartz* takes the simpler approach of using an array of signers and loops to check if enough signers have approved a transaction. This makes the *Quartz* wallets' constructors cheaper (there are fewer fields and less bytecode) and makes transactions cheaper when the total number of signers is small. The advantage of the *Quartz* wallets decreases under workloads with more signers.

The disadvantage of the *Quartz* approach is its use of loops, which means gas costs for wallet transactions grow with the number of signers. The advantage, however, is that the code generated by

Quartz is flexible, because it must accommodate any valid sequence of *Quartz* operations against the group of signers and authorization checks against it, i.e., it cannot exploit domain knowledge and optimize based on assumptions of how authorized signers are added or removed over time.

7 RELATED WORK

Smart contracts have attracted immense interest in both industry and academia, making them a popular target for language design and formal methods. Additionally, software development and verification based on state machines has a long history with many interesting applications. We summarize some of the most significant related works below.

7.1 State Machine-Based Development

Quartz is most directly inspired by prior works that similarly leverage a programming abstraction based on state machines to facilitate development and systematic testing of critical software. Teapot [10] is a domain-specific language used to write state machine implementations of the cache coherence protocol. It allows the programmer to translate their state machine into a specification for the Murphi model checker and into an implementation in C. Transit [57] takes this a step further, allowing the programmer to partially implement a distributed protocol as a state machine, then synthesizing the rest of the implementation using a counter-example guided inductive synthesis (CEGIS) loop. One could imagine extending *Quartz* to synthesize contract code using TLC in a similar approach.

Quartz is also similar to P [15], a domain-specific language for event-driven programming. Programs in P are written as state machines which can then be model checked and converted into implementations expressed in C. However, P expects the developer to formalize the state machine's environment by writing a second "ghost" state machine. *Quartz* has no similar expectation. It generates TLA⁺ code that formalizes blockchain execution semantics.

There are also works that apply state machines to smart contract development. In FSolidM [36], a user builds a contract state machine in a graphical editor, but there is no effort at verification, and the user must manually write Solidity to complete the contract. VeriSolid [37] extended FSolidM with verification, but the user must still manually write Solidity, and VeriSolid can only reason about limited properties. Finally, Obsidian [11, 12] is a contract programming language that also features state machines as the primary abstraction. It does not involve any verification, but rather uses language features like linear types to guarantee certain properties.

7.2 Contract Programming Languages

Smart contracts have become a very popular domain for new programming languages, particularly as shortcomings to Solidity [18] have emerged. Most of these are fully-featured programming languages [1, 9, 19, 28, 47, 48], without associated tools to validate contract properties. Tezos has introduced a high-level programming language [53] that compiles to a stack-based language [52] designed to be amenable to formal analysis. However, this analysis would not be automated, but rather requires manual use of a theorem prover like Coq.



Figure 8: Gas costs when executing equivalent generated and handwritten Solidity code

There are also some examples of more simplified, domain-specific contract languages. Frantz and Nowostawski [22] built a tool in which developers express the operations and rules for use of a contract in a human-readable DSL. This contract description is then translated into a Solidity template that the developer must manually complete. Findel [6] is a DSL specifically for expressing financial derivative contracts, heavily inspired by prior work on similar DSLs [46]. Finally, Cardano has built *Marlowe* [29], a Haskell-embedded DSL for financial contracts.

Quartz has similarities to *Scilla* [50], a functional style intermediate language for smart contracts. A *Scilla* contract is also written as a state machine. Unlike *Quartz, Scilla* is a fully-featured programming language rather than a DSL, and therefore tends to be less concise. Its authors report that ERC-20 and ERC-721 respectively require 158 and 270 lines of code, where *Quartz* requires 43 and 48 lines, respectively. Finally, *Scilla* does support automated contract analysis, but only through a library of pre-written static analyzers that must generalize to any contract. Unlike in *Quartz*, checking functional properties specific to a contract remains manual.

7.3 Contract Analysis Tools

While there have been several one-off efforts to verify the properties of a single contract [5, 39], many reusable tools have been built to analyze the behavior and identify vulnerabilities of existing contracts, most commonly by applying symbolic execution techniques to their EVM bytecode representations [33, 35, 38, 42] or static analysis of the contract's AST [54]. These tools offer fully automated analysis, but they are largely restricted to identifying a fixed collection of generic vulnerabilities. ZEUS [31] is a tool to convert a Solidity contract into LLVM bytecode for model checking. Securify [56] analyzes EVM bytecode to extract control flow and data flow graphs. It then verifies contract properties written in a datalog-based DSL, although these properties are intended to be generalized across many contracts rather than customized to a specific contract as in *Quartz*. VERX [45] is a verification tool for Ethereum contracts that uses predicate abstraction and symbolic execution of EVM bytecode. Like *Quartz*, it offers automated verification of contract-specific properties, written in a variant of linear temporal logic. This makes VERX arguably the most comparable analysis tool to *Quartz*. However, VERX makes the assumption that all of the contracts it analyzes are effectively callback free [23], meaning it cannot model how something like a token send may disrupt a contract, either through reentrancy or exceptions. *Quartz* does not have this restriction, meaning it could find violations to properties p_1 and p_2 discussed in Section 6.2 that VERX cannot.

8 CONCLUSION

This paper presented Quartz, a tool for implementing and testing secure smart contracts, using state machines as its fundamental organizing principle. Quartz offers developers a small, specialized language for writing contract logic with features motivated by real smart contract applications, such as transition authorization as a first-class primitive. Quartz also supports validation of a contract's properties through translation into a TLA⁺ representation suitable for model checking. Once validated, a developer may seamlessly deploy her contract to a blockchain by using Quartz to generate Solidity. Our evaluation demonstrates that Quartz contracts are more concise than their handwritten Solidity counterparts. We presented two in-depth case studies of contract validation in which Quartz surfaces significant contract vulnerabilities and offers helpful execution traces that assist the developer in patching these vulnerabilities. Finally, we have shown that Quartz imposes only modest overhead in terms of both the size and execution efficiency of its generated contract implementations.

We believe *Quartz* can serve as a platform for true smart contract engineering, replacing the more ad-hoc development and testing processes currently in use. Even well-crafted unit test suites may overlook critical contract vulnerabilities that only arise as the result of specific, unanticipated sequences of events. Through *Quartz*facilitated model checking, such execution traces are revealed and used to inform refinements to the contract's code that harden it against attack. An implementation is then generated and deployed for production use against real users and potential adversaries only once the contract is validated with respect to the desired properties.

There are a number of potential directions for future work. First, while *Quartz* state machines may currently only communicate by sending tokens, we plan to extend the DSL to support invocation of transitions in external state machines. This would require changes to TLA⁺ specification generation, but would allow developers to create blockchain applications from compositions of *Quartz* state machines. Additionally, we are considering targeting additional smart contract platforms, such as Hyperledger Fabric [2], for the deployment of *Quartz* contracts. This would involve both targeting a new contract implementation language, such as Go, and properly formalizing the execution semantics of the new platform in TLA⁺.

REFERENCES

- Aeternity. 2020. The Sophia Language. https://github.com/aeternity/aesophia/ blob/lima/docs/sophia.md.
- [2] Elli Androulaki, Artem Barger, Vita Bortnikov, Christian Cachin, Konstantinos Christidis, Angelo De Caro, David Enyeart, Christopher Ferris, Gennady Laventman, Yacov Manevich, Srinivasan Muralidharan, Chet Murthy, Binh Nguyen, Manish Sethi, Gari Singh, Keith Smith, Alessandro Sorniotti, Chrysoula Stathakopoulou, Marko Vukolić, Sharon Weed Cocco, and Jason Yellick. 2018. Hyperledger Fabric: A Distributed Operating System for Permissioned Blockchains. In Proceedings of the Thirteenth EuroSys Conference (Porto, Portugal) (EuroSys '18). ACM, New York, NY, USA, Article 30, 15 pages. https://doi.org/10.1145/ 3190508.3190538
- [3] Nicola Atzei, Massimo Bartoletti, and Tiziana Cimoli. 2017. A Survey of Attacks on Ethereum Smart Contracts. In *Principles of Security and Trust.* Lecture Notes in Computer Science, Vol. 10204. Springer, 164–186.
- [4] Massimo Bartoletti and Livio Pompianu. 2017. An Empirical Analysis of Smart Contracts: Platforms, Applications, and Design Patterns. In Financial Cryptography and Data Security. Springer, 494–509.
- [5] Giancarlo Bigi, Andrea Bracciali, Giovanni Meacci, and Emilio Tuosto. 2015. Validation of decentralised smart contracts through game theory and formal methods. In *Programming Languages with Applications to Biology and Security*. Springer, 142–161.
- [6] Alex Biryukov, Dmitry Khovratovich, and Sergei Tikhomirov. 2017. Findel: Secure Derivative Contracts for Ethereum. In *Financial Cryptography and Data Security*. Springer, 453–467.
- [7] Matthew Black and Tony Cai. 2019. ERC-1850 Hashed Time-Locked Principal Contract. https://github.com/ethereum/EIPs/pull/1850.
- [8] Matthew Black and TingWei Liu. 2018. ERC-1630 Hashed Time-Locked Contracts. https://github.com/ethereum/EIPs/issues/1631.
- [9] Sam Blackshear, Evan Cheng, David L Dill, Victor Gao, Ben Maurer, Todd Nowacki, Alistair Pott, Shaz Qadeer, Dario Russi, Stephane Sezer, Tim Zakian, and Runtian Zhou. 2020. Move: A Language With Programmable Resources. (2020), 26. https://developers.libra.org/docs/assets/papers/libra-move-a-language-withprogrammable-resources.pdf
- [10] Satish Chandra, Brad Richards, and James R Larus. 1999. Teapot: A domainspecific language for writing cache coherence protocols. *IEEE Transactions on Software Engineering* 25, 3 (1999), 317–333.
- [11] Michael Coblenz. 2017. Obsidian: A Safer Blockchain Programming Language. In Proceedings of the 39th International Conference on Software Engineering Companion (ICSE-C '17). ACM, 97–99.
- [12] Michael J. Coblenz, Jonathan Aldrich, Joshua Sunshine, and Brad A. Myers. 2018. User-Centered Design of Permissions, Typestate, and Ownership in the Obsidian Blockchain Language. In HCI for Blockchain: Studying, Designing, Critiquing and Envisioning Distributed Ledger Technologies Workshop at CHI 2018. ACM.
- [13] Michael del Castillo. 2016. The DAO Attacked: Code Issue Leads to \$60 Million Ether Theft. https://www.coindesk.com/dao-attacked-code-issue-leads-60million-ether-theft.
- [14] Kevin Delmolino, Mitchell Arnett, Ahmed Kosba, Andrew Miller, and Elaine Shi. 2016. Step by step towards creating a safe smart contract: Lessons and insights from a cryptocurrency lab. In *International Conference on Financial Cryptography* and Data Security. Springer, 79–94.
- [15] Ankush Desai, Vivek Gupta, Ethan Jackson, Shaz Qadeer, Sriram Rajamani, and Damien Zufferey. 2013. P: Safe Asynchronous Event-driven Programming. In Proceedings of the 34th ACM SIGPLAN Conference on Programming Language Design and Implementation (Seattle, Washington, USA) (PLDI '13). ACM, New

York, NY, USA, 321-332. https://doi.org/10.1145/2491956.2462184

- [16] William Entriken, Dieter Shirley, Jacob Evans, and Nastassia Sachs. 2018. ERC-721 Non-Fungible Token Standard. https://eips.ethereum.org/EIPS/eip-721.
- [17] Ethereum Foundation. 2019. Common Patterns Solidity Documentation. https://solidity.readthedocs.io/en/latest/common-patterns.html.
- [18] Ethereum Foundation. 2019. Solidity. https://solidity.readthedocs.io/en/latest/.
- [19] Ethereum Foundation. 2019. Vyper. https://github.com/ethereum/vyper.
- [20] Ethereum Foundation. 2020. Ethereum Flavored Web Assembly. https://github. com/ewasm.
- [21] Ethereum Foundation. 2020. Solidity by Example: Simple Open Auction. https://solidity.readthedocs.io/en/v0.6.6/solidity-by-example.html.
- [22] C. K. Frantz and M. Nowostawski. 2016. From Institutions to Code: Towards Automated Generation of Smart Contracts. In 2016 IEEE First International Workshops on Foundations and Applications of Self* Systems (FAS*W). IEEE, 210–215.
- [23] Shelly Grossman, Ittai Abraham, Guy Golan-Gueta, Yan Michalevsky, Noam Rinetzky, Mooly Sagiv, and Yoni Zohar. 2017. Online Detection of Effectively Callback Free Objects with Applications to Smart Contracts. Proc. ACM Program. Lang. 2, POPL, Article 48 (Dec. 2017), 28 pages. https://doi.org/10.1145/3158136
- [24] Arie Gurfinkel, Temesghen Kahsai, Anvesh Komuravelli, and Jorge A. Navas. 2015. The SeaHorn Verification Framework. In *Computer Aided Verification*, Daniel Kroening and Corina S. Păsăreanu (Eds.). Springer International Publishing, Cham, 343–361.
- [25] Hashfuture Inc. 2018. Etherescan: ERC-1540 Contract. https://etherscan.io/ address/0x565b7bd8056322f96dac28345245aead44f24ff2#code.
- [26] Thomas A. Henzinger, Ranjit Jhala, Rupak Majumdar, and Grégoire Sutre. 2003. Software Verification with BLAST. In *Model Checking Software*, Thomas Ball and Sriram K. Rajamani (Eds.). Springer Berlin Heidelberg, Berlin, Heidelberg, 235–239.
- [27] Maurice Herlihy. 2018. Atomic Cross-Chain Swaps. In Proceedings of the 2018 ACM symposium on principles of distributed computing. 245–254.
- [28] IOHK Foundation. 2018. Plutus Introduction. https://cardanodocs.com/technical/ plutus/introduction.
- [29] IOHK Foundation. 2020. Marlowe: A Contract Language for the Financial World. https://testnets.cardano.org/en/marlowe/.
- [30] Axe Jiang, Yinghao Jia, Yong Ren, and Jiaqing Dong. 2018. ERC-1540 Asset Token Standard. https://github.com/ethereum/EIPs/pull/1540.
- [31] Sukrit Kalra, Seep Goel, Mohan Dhawan, and Subodh Sharma. 2018. Zeus: Analyzing safety of smart contracts. In 25th Annual Network and Distributed System Security Symposium, NDSS. 18–21.
- [32] Igor Konnov, Jure Kukovec, and Thanh-Hai Tran. 2019. TLA+ Model Checking Made Symbolic. Proc. ACM Program. Lang. 3, OOPSLA, Article 123 (Oct. 2019), 30 pages. https://doi.org/10.1145/3360549
- [33] Johannes Krupp and Christian Rossow. 2018. teether: Gnawing at ethereum to automatically exploit smart contracts. In 27th {USENIX} Security Symposium ({USENIX} Security 18). 1317–1333.
- [34] Leslie Lamport. 2002. Specifying Systems: the TLA+ Language and Tools for Hardware and Software Engineers. Addison-Wesley Longman Publishing Co., Inc.
- [35] Loi Luu, Duc Hiep Chu, Hrishi Olickel, Prateek Saxena, and Aquinas Hobor. 2016. Making Smart Contracts Smarter. In Proceedings of the 2016 ACM SIGSAC Conference on Computer and Communications Security (Vienna, Austria) (CCS '16). ACM, 254–269.
- [36] Anastasia Mavridou and Aron Laszka. 2018. Designing secure ethereum smart contracts: A finite state machine based approach. In *International Conference on Financial Cryptography and Data Security*. Springer, 523–540.
- [37] Anastasia Mavridou, Aron Laszka, Emmanouela Stachtiari, and Abhishek Dubey. 2019. VeriSolid: Correct-by-design smart contracts for Ethereum. In International Conference on Financial Cryptography and Data Security. Springer, 446–465.
- [38] M.Mossberg, F.Manzano, E.Hennenfent, A.Groce, G.Grieco, J.Feist, T.Brunson, and A Dinaburg. 2019. Manticore: A User-Friendly Symbolic Execution Framework for Binaries and Smart Contracts. arXiv 2019. arXiv (2019). https://arxiv.org/ abs/1907.03890
- [39] Zeinab Nehai, Pierre-Yves Piriou, and Frederic Daumas. 2018. Model-checking of smart contracts. In IEEE International Conference on Blockchain. 980–987.
- [40] Chris Newcombe. 2014. Why Amazon Chose TLA + . In Abstract State Machines, Alloy, B, TLA, VDM, and Z, Yamine Ait Ameur and Klaus-Dieter Schewe (Eds.). Springer Berlin Heidelberg, Berlin, Heidelberg, 25–39.
- [41] Ivica Nikolić, Aashish Kolluri, Ilya Sergey, Prateek Saxena, and Aquinas Hobor. 2018. Finding the greedy, prodigal, and suicidal contracts at scale. In Proceedings of the 34th Annual Computer Security Applications Conference. ACM, 653–663.
- [42] Ivica Nikolić, Aashish Kolluri, Ilya Sergey, Prateek Saxena, and Aquinas Hobor. 2018. Finding the greedy, prodigal, and suicidal contracts at scale. In Proceedings of the 34th Annual Computer Security Applications Conference. ACM, 653–663.
- [43] OpenZeppelin. 2020. OpenZeppelin Contracts. https://github.com/OpenZeppelin/ openzeppelin-contracts.
- [44] Santiago Palladino. 2017. The Parity Wallet Hack Explained. https://blog.zeppelin. solutions/on-the-parity-wallet-multisig-hack-405a8c12e8f7.
- [45] Anton Permenev, Dimitar Dimitrov, Petar Tsankov, Dana Drachsler-Cohen, and Martin Vechev. 2020. Verx: Safety verification of smart contracts. In 2020 IEEE

Symposium on Security and Privacy. 17.

- [46] Simon Peyton Jones, Jean-Marc Eber, and Julian Seward. 2000. Composing Contracts: An Adventure in Financial Engineering. In Proceedings of the Fifth ACM SIGPLAN International Conference on Functional Programming (ICFP '00). ACM, New York, NY, USA, 280–292. https://doi.org/10.1145/351240.351267
- [47] Stuart Popejoy. 2017. The Pact Smart Contract Language. https://kadena.io/ docs/Kadena-PactWhitepaper.pdf
- [48] Franklin Schrans, Susan Eisenbach, and Sophia Drossopoulou. 2018. Writing Safe Smart Contracts in Flint. In Conference Companion of the 2nd International Conference on Art, Science, and Engineering of Programming (Nice, France) (Programming '18 Companion). Association for Computing Machinery, New York, NY, USA, 218–219. https://doi.org/10.1145/3191697.3213790
- [49] Ilya Sergey, Amrit Kumar, and Aquinas Hobor. 2018. Temporal Properties of Smart Contracts. In International Symposium on Leveraging Applications of Formal Methods. Springer, 323–338.
- [50] Ilya Sergey, Vaivaswatha Nagaraj, Jacob Johannsen, Amrit Kumar, Anton Trunov, and Ken Chan Guan Hao. 2019. Safer smart contract programming with Scilla. *Proceedings of the ACM on Programming Languages* 3, OOPSLA (2019), 1–30.
- [51] slock.it. 2018. DAO. https://github.com/slockit/DAO.
- [52] Tezos. 2019. The Michelson Language. https://www.michelson-lang.com/.[53] Tezos. 2020. Ligo. https://ligolang.org/.
- [54] Sergei Tikhomirov, Ekaterina Voskresenskaya, Ivan Ivanitskiy, Ramil Takhaviev, Evgeny Marchenko, and Yaroslav Alexandrov. 2018. SmartCheck: Static Analysis of Ethereum Smart Contracts. In Proceedings of the 1st International Workshop on Emerging Trends in Software Engineering for Blockchain (Gothenburg, Sweden) (WETSEB '18). Association for Computing Machinery, New York, NY, USA, 9–16. https://doi.org/10.1145/3194113.3194115
- [55] Joel Torstensson. 2017. ERC-780 Ethereum Claims Registry. https://github.com/ ethereum/EIPs/issues/780.
- [56] P. Tsankov, A. Dan, D. Drachsler-Cohen, A. Gervais, F. Bünzli, and M. Vechev. 2018. Securify: Practical Security Analysis of Smart Contracts. In Proceedings of the 2018 ACM SIGSAC Conference on Computer and Communications Security (CCS '18). 67–82.
- [57] Abhishek Udupa, Arun Raghavan, Jyotirmoy V Deshmukh, Sela Mador-Haim, Milo MK Martin, and Rajeev Alur. 2013. TRANSIT: specifying protocols with concolic snippets. ACM SIGPLAN Notices 48, 6 (2013), 287–296.
- [58] Fabian Vogelstellar and Vitalik Buterin. 2015. ERC-20 Token Standard. https: //eips.ethereum.org/EIPS/eip-20.
- [59] Zainan Victor Zhou, Evan Botello, and Yin Xu. 2018. ERC-1202 Voting Standard. https://eips.ethereum.org/EIPS/eip-1202.

A LANGUAGE SYNTAX

The syntax for *Quartz* is formally defined below. Syntax for literals as well as for arithmetic and logical operators is as expected and omitted for brevity.

```
\langle fields \rangle ::= data `{ (field \rangle * '}'
```

```
\langle field \rangle ::= name': \langle type \rangle
```

- \$\structDecl\$::= struct structName `{` \field\$* `}`
- $\langle transition \rangle ::= name': \langle sourceSt \rangle '->' \langle params \rangle dest \langle authPred \rangle \langle guard \rangle \langle transBody \rangle$

```
\langle sourceSt \rangle ::= \varepsilon | source
```

```
\langle params \rangle ::= \varepsilon | `(' \langle param-list \rangle `)'
```

```
⟨paramList⟩ ::= ⟨param⟩ `,` ⟨paramList⟩ | ⟨param⟩
```

```
\langle param \rangle ::= name `:` \langle type \rangle
```

 $\langle guard \rangle ::= \varepsilon | requires `[' \langle expr \rangle `]'$

```
\langle authPred \rangle ::= \varepsilon \mid authorized `[' \langle authExpr \rangle `]'
```

```
\langle transBody \rangle ::= `{` \langle stmt \rangle^* `}`
```

```
(type) ::= Int Uint | Timestamp | Timespan | Bool
```

```
| Map `[' \langle type \rangle `]'
```

```
| Sequence '[' \langle type \rangle ']'
```

```
| HashValue '[' (typeList) ']'
```

 $\langle typeList \rangle ::= \langle type \rangle$, $\langle typeList \rangle | \langle type \rangle$

 $\langle lValue \rangle ::= x | \langle mapRef \rangle | \langle structRef \rangle$

 $\langle mapRef \rangle ::= \langle IValue \rangle `[' \langle expr \rangle `]'$ $\langle structRef \rangle ::= \langle IValue \rangle `.' \langle expr \rangle$ $\langle expr \rangle ::= balance | sender | now | b | i | u | t | \langle IValue \rangle$ $| min `(' \langle expr \rangle `)' | max `(' \langle expr \rangle `)'$ $| size `(' \langle expr \rangle `)' | hash `(' \langle expr \rangle `)'$ $| \langle expr \rangle \langle binOp \rangle \langle expr \rangle$ $\langle authExpr \rangle ::= x | `any' x | `all' x$ $| \langle authExpr \rangle `|! ` \langle authExpr \rangle$ $| \langle authExpr \rangle `8&' \langle authExpr \rangle$ $| send \langle expr \rangle `to' \langle expr \rangle$ $| send \langle expr \rangle `to' \langle expr \rangle$ $| append \langle expr \rangle `to' \langle expr \rangle$ $| clearSeq \langle expr \rangle$ $\langle propertySpec \rangle ::= properties `{' \langle expr \rangle * '}'$

B PLUSCAL EXAMPLE

Below we provide two snippets of the PlusCal specification produced by *Quartz* for the Auction case study, with small simplifications for clarity. The original *Quartz* code was listed above in Figure 3. First, the main invocation loop is below, where any of the contract's transitions may be invoked in any order by any entity.

```
begin Main:
  with sender ∈ IDENTITIES, duration ∈ 0..MAX_INT do
      call initialize(sender, duration);
  end with:
Loop:
  either
    with sender ∈ IDENTITIES, bid ∈ 0..MAX_INT do
      call initialBid(sender, bid);
    end with:
  or
    with sender ∈ IDENTITIES, bid ∈ 0..MAX_INT do
      call submitBid(sender, bid);
  or
  or
    with sender ∈ IDENTITIES do
      call redeem(sender);
    end with;
```

```
end either;
```

PlusCal for the submtBid transition is given below, featuring checks against the current state and the transition's guard before the transition's body.

```
procedure submitBid(sender, bid)
begin submitBid:
    if currentState ≠ OPEN then
        return;
    end if;
    if bid ≤ HighestBid ∨ currentTime > Deadline then
        return;
    end if;
    balance := balance + bid;
    call send(HighestBidder, HighestBid);
    HighestBid := bid;
```

HighestBidder := sender; return; end procedure;